Author's response to reviews

Title: Microheterogeneity of transthyretin in serum and ascitic fluid of ovarian cancer patients.

Authors:

Beate Gericke (steinhag@rz.uni-potsdam.de)
Jens Raila (jraila@rz.uni-potsdam.de)
Jalid Sehouli (sehouli@charite.de)
Sophie Haebel (haebel@rz.uni-potsdam.de)
Dominique Konsgen (dkonsgen@aol.com)
Alexander Mustea (mustealex@aol.com)
Florian J Schweigert (fjschwei@rz.uni-potsdam.de)

Version: 4 Date: 4 October 2005

Author's response to reviews:

Dear editor,

the following changes took place:

References - The reference list contains now all authors' names, regardless of how many there are. The term "et al" isn't used.

Tables - These are be oriented in portrait.